Author's response to reviews

Title: Long term evolution of valsalva retinopathy: a case series

Authors:

Miriam Garcia Fernandez (migareci@hotmail.es)
Joaquín Castro Navarro (jcastro22557@gmail.com)
Carmen Gonzalez Castaño (cgcastano@hotmail.com)

Version: 5 Date: 16 August 2012

Author's response to reviews:

As requested, the manuscript has been reviewed by two different native speakers and minor changes (grammatical, fundamentally) have been done.

Many thanks,

The authors